Identifying determinants of diabetes risk and outcomes for people with severe mental illness: a mixed-methods study

Jennie Lister,¹ Lu Han,¹ Sue Bellass,² Jo Taylor,¹ Sarah L Alderson,² Tim Doran,¹ Simon Gilbody,¹ Catherine Hewitt,¹ Richard IG Holt,^{3,4} Rowena Jacobs,⁵ Charlotte EW Kitchen,¹ Stephanie L Prady,¹ John Radford,⁶ Jemimah R Ride,⁷ David Shiers,^{8,9,10} Han-I Wang¹ and Najma Siddiqi^{1,11,12*}

Declared competing interests of authors: Sarah L Alderson has received funding from the Wellcome Institutional Strategic Support Fund and National Institute for Health Research (NIHR) Clinical Trials Fellowship during the conduct of the study, and is a member of the Health Services and Delivery Research funding committee. Simon Gilbody is deputy chairperson of the NIHR Health Technology Assessment (HTA) Commissioning Board (2017 to present), and is a member of the HTA Funding Committee Policy Group (2017 to present) and the HTA Post-funding Committee teleconference (2017 to present). He was a member of the HTA Commissioning Committee (2008–17). Catherine Hewitt is a member of the NIHR HTA Commissioning Board (2015 to present). Richard IG Holt has received honoraria for speaker engagements, conference attendance or advisory boards from AstraZeneca plc (Cambridge, UK), C.H. Boehringer Sohn AG & Ko. KG (Ingelheim am Rhein, Germany), the European Association for the Study of Diabetes (Duesseldorf, Germany), Eli Lilly and Company (Indianapolis, IN, USA), Janssen Pharmaceutica (Beerse, Belgium), Menarini Group (Florence, Italy), Mylan NV (Canonsburg, PA, USA), Novo Nordisk A/S (Bagsværd, Denmark), OmniaMed Communications Ltd (London, UK) and Otsuka Pharmaceutical Co., Ltd (Tokyo, Japan), outside the submitted work. Furthermore, he was a

¹Department of Health Sciences, University of York, York, UK

²Leeds Institute of Health Sciences, University of Leeds, Leeds, UK

³Faculty of Medicine, University of Southampton, Southampton, UK

⁴University Hospital Southampton NHS Foundation Trust, Southampton, UK

⁵Centre for Health Economics, University of York, York, UK

⁶Patient and public involvement representative, Keighley, UK

⁷Centre for Health Policy, School of Population and Global Health, University of Melbourne, Melbourne, VIC, Australia

⁸Division of Psychology and Mental Health, University of Manchester, Manchester, UK

⁹Psychosis Research Unit, Greater Manchester Mental Health NHS Foundation Trust, Manchester, UK

¹⁰Primary Care and Health Sciences, Keele University, Keele, UK

¹¹Hull York Medical School, University of York, York, UK

¹²Bradford District Care NHS Foundation Trust, Bradford, UK

^{*}Corresponding author najma.siddiqi@york.ac.uk

member of the HTA Prioritisation Committee C (Mental Health, Women and Children's Health) until July 2019. David Shiers is an expert advisor to the National Institute for Health and Care Excellence (NICE) centre for guidelines and a member of the current NICE guideline development group for rehabilitation in adults with complex psychosis and related severe mental health conditions. He is also a board member of the National Collaborating Centre for Mental Health (NCCMH) and a clinical advisor (paid consultancy basis) to the National Clinical Audit of Psychosis (NCAP); these are the personal views of David Shiers and not those of NICE, NCCMH or NCAP. David Shiers has received personal fees from the publication Promoting Recovery in Early Psychosis (French P, Smith J, Shiers D, Reed M, Rayne M. *Promoting Recovery in Early Psychosis: A Practice Manual.* Chichester: Wiley-Blackwell; 2010), as a joint editor in receipt of royalties, and has received personal fees as a member of the current NICE guideline development group for rehabilitation in adults with complex psychosis and related severe mental health conditions.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published May 2021 DOI: 10.3310/hsdr09100

Plain English summary

Determinants of diabetes risk for people with severe mental illness

Health Services and Delivery Research 2021; Vol. 9: No. 10

DOI: 10.3310/hsdr09100

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Severe mental illness refers to a group of illnesses, such as schizophrenia, that greatly interfere with life activities. People with severe mental illness die earlier and have worse physical health than the general population. They are more than twice as likely to develop diabetes and to get more complications from having diabetes. It is currently unclear how severe mental illness interacts with diabetes or how having both conditions influences health-care use.

We looked at general practice records from large numbers of patients across England. We also interviewed 39 people with severe mental illness and diabetes, nine family members and 30 health-care professionals across the North West, and Yorkshire and the Humber.

Key findings

In people with severe mental illness, older age, being from an ethnic minority, living in a deprived area, having multiple health conditions and using certain medications predicted the development of diabetes.

For people with both severe mental illness and diabetes, being older, living in deprived areas and having lots of different health conditions led to poorer physical and mental health outcomes.

Participants with severe mental illness said that they put dealing with their mental illness above caring for their physical health. They also struggled to manage lots of health problems.

Physical health problems were often overlooked by health-care services because physical and mental health problems are often treated in separate services. A keyworker responsible for co-ordinating care could help to bridge this gap.

Good social support, diabetes knowledge and better mental health (like better mood) were things that helped people with severe mental illness to better manage their diabetes. Health-care staff wanted more training about physical or mental health problems.

Participants' low levels of physical activity and poor mental and physical health were barriers to effective diabetes management. These barriers need to be thought about when designing better treatments.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 15/70/26. The contractual start date was in September 2017. The final report began editorial review in December 2019 and was accepted for publication in June 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

Copyright © 2021 Lister *et al.* This work was produced by Lister *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk